Author's response to reviews

Title: Urological research in sub-Saharan Africa: a retrospective cohort study of abstracts presented at the Nigerian Association of Urological Surgeons Conferences

Authors:

Jibril O Bello (jabarng@yahoo.com)

Version: 3 Date: 10 November 2013

Author's response to reviews:

The Editorial Office,
BMC Urology.

COVER LETTER FOR MANUSCRIPT (MS 4451052168842312) TITLED : Fate of abstracts presented at the Nigerian Association of Urological Surgeons Conferences: a retrospective cohort study.

The reviewers' comments are noted and corrections have been made to the manuscript.
Tables 2 and 3 have been grouped together and its presentation of result data improved on.
A discussion on the period of observation for publications has been included in the discussion section.
The legend for Figure 2 has been improved and now includes a note on the databases searched and period of observation for publication.
Discretionary revision - changes in first authorship between presented abstract and subsequent published manuscript was not analyzed.
Thank you both for the encouraging comments.

Bello, Jibril Oyekunle (Corresponding Author).